Author's response to reviews

Title: Primary parotid gland lymphoma: A case report

Authors:

Petros Konofaos (petros_konofaos@yahoo.com)
Eleftherios Spartalis (eleftherios.spartalis@gmail.com)
Paraskevas Katsaronis (pkatsaronis@yahoo.gr)
Grigoris Kouraklis (gkouraklis@hotmail.com)

Version: 3  Date: 26 January 2011

Author's response to reviews: see over
Responses to Reviewer

1. Abstract:
   Line 5: "and a brief review of the literature" should be omitted.
   Omitted.

2. Case presentation:
   Line 1: "Greek" should be replaced by "caucasian"
   Replaced.

3. Figure 2:
   The patient's eyes must be covered
   Done.